Author's response to reviews

Title: The effect of participatory women's groups on birth outcomes in Bangladesh: does coverage matter? A cluster randomized trial.

Authors:

Tanja AJ Houweling (tanja.houweling@gmail.com)
Kishwar Azad (pcp@dab-bd.org)
Layla Younes (younes.layla@gmail.com)
Abdul Kuddus (kudduspcp@dab-bd.org)
Sanjit Kumar Shah (pcp@dab-bd.org)
Bedowra Haq (pcp@dab-bd.org)
Tasmin Nahar (pcp@dab-bd.org)
James Beard (j.beard@ucl.ac.uk)
Edward F Fottrell (e.fottrell@ucl.ac.uk)
Audrey Prost (prostaudrey@gmail.com)
Anthony Costello (cihdcostello@gmail.com)

Version: 6 Date: 16 August 2011

Author's response to reviews: see over
Dear Editor,

Thank you for giving us the opportunity to submit a revised version of our manuscript “The effect of participatory women’s groups on birth outcomes in Bangladesh: does coverage matter? A cluster randomized trial” (MS: 1425426120542632).

We thank the reviewer for his/her appropriate comments, all of which have been addressed accordingly. A detailed account of all revisions and responses is given below.

We thank you for your consideration.

Yours sincerely,

Tanja AJ Houweling (corresponding author)

UCL Centre for International Health and Development,
Institute of Child Health
30 Guildford Street
London WC1N 1EH
### Response to the reviewer’s comments

<p>| | | |</p>
<table>
<thead>
<tr>
<th></th>
<th></th>
<th></th>
</tr>
</thead>
<tbody>
<tr>
<td><strong>1. P(age) 13, p(aragraph) 2.</strong> While the authors claim the names are listed alphabetically, on lines 2 and 3 the names: Mandiz, Dhakai and Pal do not appear to be in alphabetical order, as well as Prost on the last l.</td>
<td>Changed accordingly.</td>
<td></td>
</tr>
<tr>
<td><strong>2. P 14, R(eference) 16, l 2.</strong> List the date of last access like R 5 and 19.</td>
<td>Changed accordingly.</td>
<td></td>
</tr>
</tbody>
</table>